Author’s response to reviews

Title: Treatment Selection of Early Stage Non-Small Cell Lung Cancer: The Role of the Patient in Clinical Decision Making

Authors:
Sahar Mokhles (s.mokhles@erasusmc.nl)
Joost Nuyttens (j.nuyttens@erasusmc.nl)
Mark de Mol (MdeMol1@amphia.nl)
Joachim Aerts (j.aerts@erasusmc.nl)
Alex Maat (a.p.w.m.maat@erasusmc.nl)
Ozcan Birim (o.birim@erasusmc.nl)
Ad Bogers (a.j.j.c.bogers@erasusmc.nl)
Hanneke Takkenberg (j.j.m.takkenberg@erasusmc.nl)

Version: 2 Date: 22 Sep 2017

Author’s response to reviews:

Thank you for the opportunity to revise our manuscript.

According to point 6 of Editorial comments we should submit the questionnaire with the submission of the revised version. However, this is a Dutch questionnaire which we would rather not want to translate or submit. First part of the questionnaire is explained in our method section and the Dutch version of Decisional Conflict Scale is widely available via internet (and mentioned in our method section with a reference).